# Fine-Scale Phylogenetic Discordance across the House Mouse Genome

# Michael A. White<sup>1</sup>, Cécile Ané<sup>2,3</sup>, Colin N. Dewey<sup>4,5,6</sup>, Bret R. Larget<sup>2,3</sup>, Bret A. Payseur<sup>1</sup>\*

1 Laboratory of Genetics, University of Wisconsin, Madison, Wisconsin, United States of America, 2 Department of Statistics, University of Wisconsin, Madison, Wisconsin, United States of America, 3 Department of Botany, University of Wisconsin, Madison, Wisconsin, United States of America, 4 Department of Biostatistics, University of Wisconsin, Madison, Wisconsin, United States of America, 5 Department of Medical Informatics, University of Wisconsin, Madison, Wisconsin, United States of America, 6 Department of Computer Sciences, University of Wisconsin, Madison, Wisconsin, United States of America

# Abstract

Population genetic theory predicts discordance in the true phylogeny of different genomic regions when studying recently diverged species. Despite this expectation, genome-wide discordance in young species groups has rarely been statistically quantified. The house mouse subspecies group provides a model system for examining phylogenetic discordance. House mouse subspecies are recently derived, suggesting that even if there has been a simple tree-like population history, gene trees could disagree with the population history due to incomplete lineage sorting. Subspecies of house mice also hybridize in nature, raising the possibility that recent introgression might lead to additional phylogenetic discordance. Single-locus approaches have revealed support for conflicting topologies, resulting in a subspecies tree often summarized as a polytomy. To analyze phylogenetic histories on a genomic scale, we applied a recently developed method, Bayesian concordance analysis, to dense SNP data from three closely related subspecies of house mice: *Mus musculus musculus, M. m. castaneus*, and *M. m. domesticus*. We documented substantial variation in phylogenetic history across the genome. Although each of the three possible topologies was strongly supported by a large number of loci, there was statistical evidence for a primary phylogenetic history in which *M. m. musculus* and *M. m. castaneus* are sister subspecies. These results underscore the importance of measuring phylogenetic discordance in other recently diverged groups using methods such as Bayesian concordance analysis, which are designed for this purpose.

Citation: White MA, Ané C, Dewey CN, Larget BR, Payseur BA (2009) Fine-Scale Phylogenetic Discordance across the House Mouse Genome. PLoS Genet 5(11): e1000729. doi:10.1371/journal.pgen.1000729

Editor: Mikkel H. Schierup, University of Aarhus, Denmark

Received April 8, 2009; Accepted October 19, 2009; Published November 20, 2009

**Copyright:** © 2009 White et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** MAW was supported by an NLM graduate training grant (NLM 2T15LM007359) to the Computation and Informatics in Biology and Medicine Training Program at the University of Wisconsin. The research was supported by an NSF grant (DEB 0918000) to BAP. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

# Introduction

With the advent of new sequencing technologies, the reconstruction of phylogenetic histories on the genomic scale has become feasible. Genomic data offer the potential to resolve phylogenies that have been difficult to reconstruct from a small number of genes [1-5]. Although highly resolved phylogenies can sometimes be recovered when data sets are concatenated, such "total evidence" trees may depart from the history of population branching, the "species history" [6,7]. The measurement and incorporation of gene genealogical discordance into genomic analyses is expected to improve inferences about species history, particularly for recently derived groups [8].

Topological discordance among gene trees is expected under several scenarios [9]. Population subdivision and asymmetric gene flow among ancestral populations [10], as well as introgression between diverged populations, can generate widespread discordance. Ancestral polymorphisms can also segregate, causing some gene trees to disagree with the population tree. The effects of this incomplete lineage sorting are greatest when effective population sizes are high and internodes of the population tree are of short duration [11–16]. Consistent with these predictions, substantial phylogenetic discordance has been documented on the genomic scale in a few young species groups. Pollard et al. [17] demonstrated significant variation among 9,405 genes in *Drosophila erecta*, *D. melanogaster*, and *D. yakuba*. In addition, genomic discordance has been repeatedly observed in analyses of humans, chimpanzees, and gorillas, with a majority of gene trees supporting a human/chimpanzee sister relationship [18–25]. Although it is well established that closely related lineages will exhibit substantial genealogical discordance, few studies have quantified phylogenetic discordance across entire genomes (including non-coding regions). Consequently, the extent of variation on this scale remains poorly understood.

The house mouse subspecies group (*Mus musculus musculus*, *M. m. castaneus*, and *M. m. domesticus*) provides an excellent system for exploring genome-wide patterns of phylogenetic discordance because (i) sources of potential discordance (incomplete lineage sorting and introgression) exist and (ii) almost complete genome sequences are available. The earliest divergences in the house mouse subspecies group occurred only 500,000 generations ago (assuming 1 generation per year) [26–30] and house mice are estimated to have large effective population sizes (approximately  $10^5$ ) [30,31], suggesting an important role for incomplete lineage sorting. In addition, the extent of interspecific gene flow varies across the genome and among the three subspecies [30,32–38].

<sup>\*</sup> E-mail: payseur@wisc.edu

### **Author Summary**

The phylogenetic history of individual genes can differ strongly from the species history if taxa are recently derived, making inferences of a species history from only a handful of genes especially difficult in these cases. Genome-scale data sets now allow phylogenetic histories to be reconstructed from a large number of genes. Although data sets of this size are becoming more common, few studies have characterized variation in phylogenetic history across whole genomes. We summarize fine scale variation in phylogenetic history across the genome of house mice, a recently derived group of subspecies, using a method that combines phylogenetic uncertainty among gene trees. We document substantial variation in phylogenetic history among 14,081 loci and describe a primary history in the face of this variation. These results support the use of genome-scale datasets and methods that accommodate phylogenetic discordance in attempts to reconstruct the history of closely related groups.

Two of the subspecies (M. m. domesticus and M. m. musculus) meet in a stable hybrid zone, in which dramatic variation in introgression among genomic regions has been documented [34,36,38,39]. The other two subspecies pairs (M. m. castaneus/M. m. domesticus and M.m. musculus/M. m. castaneus) also exchange genes in nature, as evidenced by the existence of hybrids [40–43]. Furthermore, house mice are model systems for the genetics of speciation [44– 49], providing the potential to connect genomic variation in phylogenetic history to the evolution of reproductive barriers.

Previous phylogenetic analyses of house mouse subspecies have revealed signs of discordance among genomic regions. Analyses of mitochondrial genes [50,51], and studies of a handful of genes on the Y chromosome [52,53], the X chromosome [30], and the autosomes [30,53,54] have yielded support for a *M. m. musculus/M. m. castaneus* sister relationship. In contrast, complete mitochondrial genome sequences have been unable to resolve the branching pattern [55]. As an increasing number of loci and individuals (within subspecies) have been surveyed, greater evidence for conflicting (but individually well-supported) topologies and nonmonophyletic clades has surfaced [30,53,54].

The discordance observed in these small datasets motivated us to characterize how frequently gene genealogies fluctuate across the genome and whether a primary phylogenetic history can be identified among the discordance. The accuracy of species tree inference is improved more by increasing the number of loci sampled than by increasing the number of individuals sampled at each locus [12]. In this three-taxon case, the primary phylogenetic history should follow the subspecies tree [56]. We applied a recently developed analytical approach designed to measure and incorporate phylogenetic discordance to genome sequences from *M. m. musculus*, *M. m. castaneus*, and *M. m. domesticus*. We document substantial, fine-

scale discordance among genomic regions and report a primary phylogenetic history for house mice supported by a plurality of the genome. We interpret these results in the context of population genetic processes, including speciation, in house mice.

# Results

# Species and Sequence Data

To reconstruct the phylogenetic history of house mice, we analyzed genome sequences of three wild-derived inbred strains obtained by Perlegen Sciences using high-density 25-mer oligonucleotide arrays [57]: WSB/EiJ (M. m. domesticus), PWD/PhJ (M. m. musculus), and CAST/EiJ (M. m. castaneus). With three subspecies, there were three possible rooted topologies (Figure 1). We identified assays within the Perlegen data set that had a single nucleotide polymorphism (SNP) between one of the three strains. For each assay, we replaced the C57BL/6J genotype with the strain-specific genotype in the mouse genome [58] using positional information from NCBI build 36, to create a unique genome sequence for each strain. Between the three inbred strains, there were a total of 4,359,927 SNPs genome-wide, with an average of 2.35 SNPs/kb. Estimates from natural populations of the three species of mice have revealed an average divergence among a handful of loci of about 5 SNPs/kb (in all three pairwise species comparisons) [30]. The lower SNP density in this data set reflects efforts by Perlegen to minimize the false positive rate at the expense of a high false negative rate during resequencing [57,59]. We used the *Rattus norvegicus* genome sequence [58] as an outgroup in all phylogenetic analyses. Rat, which diverged from house mice 12-24 million years ago [60,61], was the most closely related species with a complete genome sequence available. The entire data matrix contained a total of 1,085,916 parsimony phylogenetic informative sites across the genome, with an average of 604 informative sites per Mb.

#### Genome Partitioning

We partitioned the genome for subsequent phylogenetic analyses using the minimum description length (MDL) principle [62], which set breakpoints where shifts in phylogenetic history most likely occurred. In this manner, the genome was partitioned into 14,081 loci with a median size of 98,238 bp and a maximum locus size of 7.21 Mb (Figure S1). Locus size varied widely across the genome (SD 312,637 bp) and was negatively correlated with the density of parsimony phylogenetically informative sites (Spearman's rank correlation, rho = -0.501, p < 0.00001), as expected because the genome was partitioned based upon 100 SNP windows rather than windows of constant physical position. In regions of lower SNP density, 100 SNP windows encompassed larger stretches of the genome.

To determine whether the distribution of locus sizes correlated with relevant biological processes (rather than reflecting an arbitrary partitioning based on the density of informative sites), we compared locus size to fine-scale recombination rate across the



Figure 1. Three possible phylogenetic histories. The three possible phylogenetic relationships among subspecies of house mice are shown, rooted by rat. doi:10.1371/journal.pgen.1000729.g001

PLoS Genetics | www.plosgenetics.org

genome. Three-species models predict that the spatial scale of phylogenetic switching caused by incomplete lineage sorting should be related to the local rate of recombination [63] and consequent scale of linkage disequilibrium in ancestral populations. This pattern has been observed in phylogenomic studies of humans, chimpanzees, and gorillas [24]. We estimated recombination rate in 1 Mb windows using an updated version of the high-density mouse genetic map [64,65] and used these estimates to examine correlations with locus size across the genomes of the three species. As predicted, we found a low, but significant negative correlation between recombination rate and locus size (Spearman's Rank, n = 14,081, rho = -0.0632, p < 0.00001).

Because marker density in the genetic map varies across the genome, the closest estimate of recombination rate can also vary. We repeated the correlation analysis using only loci that were within 100 kb of the nearest recombination rate estimate, resulting in a similar negative correlation (Spearman's Rank, n = 8,344, rho = -0.0744, p<0.00001). We also compared median recombination rates between the longest 2.5% and the shortest 2.5% of the loci. There was a significantly lower median recombination rate in the longer loci (0.292 cM/Mb) as compared to the shorter loci (0.480 cM/Mb; p<0.00001) (Figure S2), confirming an effect of recombination on the spatial distribution of phylogenetic discordance.

# Estimating Single-Locus Phylogenetic Histories

We separately estimated the rooted gene genealogy of each locus by identifying the best-fitting model of molecular evolution and conducting a Bayesian phylogenetic analysis. These individual loci supported single topologies with high statistical confidence (Table S1). For example, 84.9% of loci supported a single topology with a posterior probability of 0.9 or greater (Figure S3). High posterior probabilities were observed across a wide variation of locus sizes and for each topology (Figure 2; Figure S4). Resolved gene trees can be supported by artificially high Bayesian posterior probabilities if the true tree is a polytomy (the "star tree paradox") [66-69]. To evaluate whether the high support we obtained for each locus was caused by this problem, we calculated the likelihood score of the Bayesian majority rule consensus tree and the likelihood score of the tree with the internal branch constrained to length zero. We computed a likelihood ratio test statistic to determine if the model with an internal branch of nonzero length was a significantly better fit than a model with an internal branch of zero length. Only 247 of the 14,081 loci had internal branches statistically indistinguishable from a tree with a collapsed internal branch (significance set at  $\alpha = 0.1$ , 1 df,  $\chi^2 = 2.706$  [70], indicating that the high posterior probabilities we observed generally reflected the accurate resolution of gene trees with short internal branches.

We compared the single-locus posterior probabilities we estimated through the Bayesian phylogenetic analyses with bootstrap support from maximum likelihood analyses to determine if the high support we obtained was due to the choice of methodology. Bootstrap supports from the maximum likelihood analyses were highly correlated with the Bayesian posterior probabilities for each of the three topologies (*M. m. musculus/M. m.*).



**Figure 2. Fine-scale phylogenetic discordance.** The posterior probability of each topology is mapped along chromosome 19 to characterize fine-scale patterns of discordance among the 410 loci. Many regions of the chromosome rapidly switch between phylogenetic histories and are characterized by loci that have a high posterior probability for a single topology. Colors correspond to the three topologies. doi:10.1371/journal.pgen.1000729.g002

*castaneus*: Spearman's Rank, n = 14,081, rho = 0.928, p < 0.00001; *M. m. castaneus/M. m. domesticus*: rho = 0.925, p < 0.00001; *M. m. musculus/M. m. domesticus*: rho = 0.880, p < 0.00001), indicating that overall patterns of statistical support at each locus were robust to the method used to reconstruct phylogenetic history.

#### Genome-Wide Discordance

To measure phylogenetic discordance across all loci, we used the posterior probability distributions from all single-locus phylogenetic analyses as input for the Bayesian concordance analysis [71]. By incorporating the statistical uncertainty in phylogenetic reconstruction among the individual loci, we were able to estimate the number of loci across the genome that supported each individual topology. Bayesian concordance analysis identified a primary phylogenetic history, placing M. m. musculus and M. m. castaneus as sister subspecies (Figure 3). This tree was supported by a concordance factor of  $0.390 \pm 0.003$ , or 39% of all loci (with a prior probability of gene tree concordance at  $\alpha = 1$ ). There was also substantial support for the two other possible histories. The M. m. castaneus/M. m. domesticus clade had a concordance factor of  $0.363 \pm 0.003$ , while the *M. m. musculus/M*. *m. domesticus* clade had a lower concordance factor of  $0.247 \pm 0.003$ . Although the Bayesian concordance analysis estimates the proportion of loci supporting a particular topology, the analysis does not integrate the sizes of individual loci. If the concordance factors are accurately reflecting the contributions of incomplete lineage sorting and gene flow, the median locus sizes supporting each of the three topologies should parallel the concordance factors. Median locus size followed the same rank order as the concordance factors, with the M. m. musculus/M. m. castaneus topology having the largest locus size (Figure S5).

Other researchers have computed concordance factors using maximum likelihood methods by filtering out gene trees with less than 95% bootstrap support and calculating the proportion of loci from the filtered pool that supported each topology [23]. We also repeated our concordance analysis using maximum likelihood methods. 7,260 of the 14,081 loci had at least 95% bootstrap support. Of these trees, *M. m. musculus/M. m. castaneus* was supported by a concordance factor of 0.415, *M. m. castaneus/M. m. domesticus* had a concordance factor of 0.366, and *M. m. musculus/M. m. domesticus* had a concordance factor of 0.219. These concordance factors were similar to those resulting from Bayesian concordance analysis, indicating that our conclusions were robust to alternative analytical approaches. The slight differences likely arose because Bayesian concordance analysis used all loci and incorporated uncertainty across loci, whereas the maximum likelihood method only used half of the available loci and did not propagate uncertainty across loci.

Bayesian concordance analysis uses a prior probability of gene tree concordance, which could affect our estimates of a primary phylogenetic history. To address this issue, we recalculated concordance factors assuming two extreme priors: (1) a high probability of concordance among individual loci, and (2) complete independence among individual loci. In both cases, patterns of discordance among the three histories remained largely unchanged (Figure 3), suggesting robustness to prior assumptions. This robustness was likely due to the large number of loci used in the concordance analysis combined with the high degree of support for single topologies at most individual loci.

Estimates of concordance might also be affected by the parameters used in the MDL partitioning (the cost and the starting interval size). To investigate this possibility, we applied the maximum (3) and minimum (0.9039) costs against splitting concatenated fragments in the MDL partitioning on chromosomes 18, 19, and X. In all cases, partitioning the genome with the minimum cost roughly doubled the number of loci on each chromosome, but chromosome-wide concordance factors were not significantly altered (Figure S6). We also calculated concordance factors on chromosomes 18 and 19 using a range of starting interval sizes (25, 50, 100, 150, 200, and 250 SNPs). For both chromosomes, concordance factors did not significantly differ for



**Figure 3. Genome-wide phylogenetic discordance.** Bayesian concordance factors were calculated from the posterior probability distributions of 14,081 single-locus phylogenetic analyses. The concordance analysis is depicted using three different priors on gene tree concordance: complete independence among loci ( $\alpha = infinity$ ), a high probability of concordance among loci ( $\alpha = 10^{-6}$ ), and an intermediate level of concordance ( $\alpha = 1$ ). The *M. m. musculus/M. m. castaneus* topology is supported by significantly more loci than the other two topologies regardless of the prior on gene tree concordance. Colors correspond to the three topologies. Error bars are 95% credibility intervals. doi:10.1371/journal.pgen.1000729.g003

the three topologies with starting interval sizes of 25, 50, or 100 SNPs when the full range of credibility intervals were taken into account (Figure S7), indicating that partitioning with starting intervals smaller than 100 SNPs did not significantly alter the estimates of concordance.

The phylogenetic histories at individual loci may not be the true histories if the divergence time between rat and house mice is too great [72,73]. Under this scenario, the rat branch can pair with whichever mouse lineage has the greatest amount of divergence and the largest number of sequence similarities due to homoplasy rather than orthology (long-branch attraction). To evaluate this possibility, we randomized the nucleotides of the rat sequence at each locus, erasing any phylogenetic signal and further compounding the effect of long-branch attraction. The posterior probability distributions from each shuffled locus on chromosomes 18 and 19 were used as input for the Bayesian concordance analysis [71]. If the patterns we observed in the data were due to long-branch attraction, we would expect to recover similar patterns of discordance with an artificially lengthened branch. Instead, we found a large difference in discordance between the randomized and nonrandomized data sets, with the randomized data set not differing from concordance factors of 1/3 (chromosome 18:  $\chi^2 = 4.900$ , df = 2, p = 0.086; chromosome 19:  $\chi^2 = 3.286$ , df = 2, p = 0.193) (Figure S8). This indicated that the rat sequence provided strong phylogenetic signal.

#### Patterns of Discordance Within the X Chromosome

Patterns of phylogenetic discordance within the X chromosome are expected to differ from those on the autosomes. The X chromosome has a smaller effective population size than the autosomes (<sup>3</sup>/<sub>4</sub> as large, assuming a breeding sex ratio of one) leading to the prediction that ancestral polymorphism should sort more quickly. Additionally, loci on the X chromosome exhibit reduced gene flow within and between species of house mice [30,32,33,36,38,39,74,75]. Both factors should reduce discordance across the X chromosome. In agreement with patterns for the autosomes, the primary phylogenetic history of the X chromosome was a M. m. musculus/M. m. castaneus sister relationship. As predicted, this history was supported by a higher concordance factor  $(0.450 \pm 0.018)$  on the X chromosome than on the autosomes (Figure 4). In addition, loci supporting a M. m. castaneus/M. m. domesticus topology had a lower concordance factor  $(0.324\pm0.018)$  on the X chromosome than on the autosomes. Although the concordance factor supporting a M. m. musculus/M. m. domesticus topology  $(0.226 \pm 0.016)$  was also lower than the autosomes, the 95% credibility intervals overlapped. Median locus sizes matching each topology on the X chromosome also paralleled the concordance factors (as observed on the autosomes), with the M. m. musculus/M. m. castaneus topology showing a larger deviation in size from the two minor topologies (Figure S5).

Increased support for the *M. m. musculus/M. m. castaneus* primary phylogenetic history and decreased support for the *M. m. castaneus/ M. m. domesticus* minor history indicated reduced phylogenetic discordance on the X chromosome. To determine whether the reduced discordance simply arose from differences in sample size ( $n_X = 442$  loci,  $n_{autosomes} = 13,639$  loci), we compared the concordance factors on the X chromosome to concordance factors estimated from sets of 442 randomly drawn loci from the autosomes (5000 replicates). The reduced discordance on the X chromosome persisted in these comparisons (*M. m. musculus/M. m. castaneus*: higher, p = 0.0062; *M. m. castaneus/M. m. domesticus*: lower, p = 0.038; *M. m. musculus/M. m. domesticus*: no difference, p = 0.326). Reduced discordance on the X chromosome thus appears to



**Figure 4. Phylogenetic discordance between the autosomes and the X chromosome.** Significantly less discordance is observed across loci of the X chromosome (X: 442 loci) than the autosomes (A: 13,639 loci). This is shown by significantly higher support for the primary phylogenetic history, the *M. m. musculus/M. m. castaneus* topology, at the expense of loci supporting the other two topologies. Colors correspond to the three topologies. Error bars are 95% credibility intervals. doi:10.1371/journal.pgen.1000729.g004

reflect processes differentially affecting the X chromosome and the autosomes.

# Ascertainment Bias in SNP Discovery

A potential source of discordance in our results comes from ascertainment bias in SNP identification. Sequences for the three house mouse species were obtained using arrays designed from a C57BL/6J reference sequence, which assayed for SNPs from the reference sequence by hybridizing oligonucleotide probes. Strong sequence divergence from the reference sequence could result in inefficient hybridization of the probes, increasing the false negative rate by incorrectly calling a C57BL/6J genotype. Consistent with such a bias, a deficiency of SNPs from the three species was documented [59]. The strongest bias was against M. m. castaneus specific SNPs - SNPs at which the M. m. castaneus strain differs from the M. m. domesticus and M. m. musculus strain. This deficiency could reduce the number of loci supporting a M. m. musculus/M. m. domesticus sister relationship (consistent with our results). To determine whether ascertainment bias would affect our ability to resolve a primary phylogenetic history, we simulated increased ascertainment bias against M. m. castaneus. M. m. castaneus informative SNPs were randomly removed from each locus across chromosomes 18 and 19 at varying levels of severity (ranging from 10-80% removed) by converting the M. m. castaneus specific SNP to the C57BL/6J genotype. All phylogenetic analyses were then repeated. These simulations modeled the effects of artificially increasing the false negative rate of SNP identification against M. m. castaneus. For each chromosome, as ascertainment bias against M. m. castaneus was increased, the concordance factor supporting the M. m. musculus/M. m. domesticus topology decreased linearly (Figure 5). Importantly, introducing ascertainment bias did not differentially affect the inferred concordance factors for the other two topologies; both factors increased at equal rates. Although ascertainment bias against M. m. castaneus SNPs existed in this data set, our main conclusions of a M. m. musculus/M. m. castaneus primary phylogenetic history and substantial phylogenetic discordance across the genome were mostly unaffected by this bias.

A second form of bias could also increase the branch length of M. m. domesticus in relation to the M. m. musculus and M. m. castaneus

branches. Perlegen discarded SNPs that were polymorphic in only one of the 15 strains sequenced. As a majority of the strains were *M. m. domesticus* in origin [57,59], the analysis favored SNP discovery in this subspecies. If this bias increased the branch length of *M. m. domesticus* across the genome, this could raise support for the primary *M. m. musculus/M. m. castaneus* topology through longbranch attraction to rat. When the rat sequence was randomized, support for the *M. m. musculus/M. m. castaneus* and *M. m. castaneus/ M. m. domesticus* topologies both decreased. However, on both chromosomes, the *M. m. musculus/M. m. castaneus* topology decreased to a greater degree (Figure S8). This result indicated that the *M. m. musculus/M. m. castaneus* topology was least affected by long-branch attraction and that the high support we observed for the primary phylogenetic history came from orthology to the rat sequence rather than ascertainment bias.

#### Discussion

Our genomic analysis revealed a primary phylogenetic history across the house mouse genome, placing *M. m. musculus* and *M. m. castaneus* as sister subspecies. We also documented striking phylogenetic discordance on a genome-wide scale. Discordance was observed in previous phylogenetic studies of house mice based on a small number of loci [30,53,54]. In addition, gene trees reconstructed from large population samples have shown that reciprocal monophyly between subspecies is higher on the X chromosome than the autosomes [29–31], a result that agrees with our genomic comparisons. In summary, our results extend previous observations from phylogenetic analyses of a few loci to the entire genome, thereby providing the power needed to resolve the history of these closely related subspecies for the first time.

In the case of three taxa where there is an underlying tree-like population history (the "species tree"), the primary phylogenetic history is expected to match the species history, whereas the frequencies of the minor trees should reflect the contributions of incomplete lineage sorting and differential gene flow [9,10,12,56,76]. Several additional factors might shape the discordance we observed. We now discuss the importance of each potential source of discordance in turn.



**Figure 5. Simulated ascertainment bias against** *M. m. castaneus.* Varying proportions of *M. m. castaneus* specific SNPs were removed from chromosomes 18 and 19 to simulate increased ascertainment bias against this taxon. The difference in concordance factors between *M. m. musculus/ M. m. castaneus* and *M. m. castaneus/M. m. domesticus* do not significantly differ at all levels of artificial bias. This indicates that recovery of a *M. m. musculus/M. m. castaneus* primary phylogenetic history is robust to increased ascertainment bias. Colors correspond to the three topologies. Error bars are 95% confidence intervals. doi:10.1371/journal.pgen.1000729.g005

# Biases in Phylogenetic Reconstruction

Errors at several stages of phylogenetic reconstruction could generate phylogenetic discordance [76-78]. First, mis-estimated models of molecular evolution could introduce disagreement among loci. However, by statistically selecting the best-fitting model of molecular evolution separately for each locus, we minimized errors associated with assuming the same model across loci. Second, the alignment with rat might have inflated discordance if the error rate in the whole-genome alignment was high. Contrary to this idea, randomizing the rat sequence in respect to the three mouse sequences across chromosomes 18 and 19 strongly reduced the posterior probabilities at individual loci and instead exacerbated discordance, suggesting that the rat sequence contributed a strong phylogenetic signal. Third, estimates of concordance might be affected by the parameters used in the MDL partitioning. Applying the minimum cost against splitting concatenated fragments roughly doubled the number of loci, but concordance factors were not significantly altered. In addition, partitioning the genome using starting intervals less than 100 SNPs had no significant effect on the concordance factors. Finally, concordance factors might have been inaccurately estimated because many SNPs were missed by resequencing [57]. Although comparable analyses of complete genome sequences would likely reveal variation in the exact breakpoints of partitions, reduced numbers of informative sites did not seem to be responsible for the observed discordance. Our analyses demonstrated that MDL partitions the genome in a phylogenetically informative manner and that individual loci generally favor one history with high posterior probability. In addition, we found a significant correlation between locus size and recombination rate across the genome (as predicted by theory), suggesting that this dataset contains information about the evolutionary processes responsible for phylogenetic discordance.

Although we detected a significant negative correlation between locus size and recombination rate, the correlation coefficient was relatively low, indicating that most of the variation in locus size was explained by other variables. The weakness of this correlation was expected for several reasons. First, our data set was limited by the number of informative sites generated by the resequencing project. Additional sequence data might change the locations of breakpoints inferred by the MDL partitioning, which would alter the locus sizes and the correlation with recombination rate. Second, the recombination rate estimates came from crosses between other inbred strains of mice [64,65], not the wild-derived strains used in our analyses. Differences in recombination rate between some of the strains used to construct the mouse genetic map and one of the wild-derived strains included in our analysis (CAST/Ej] have been observed [79].

# Patterns of Discordance across the House Mouse Genome

Pairwise divergence times between house mouse subspecies pairs are roughly similar when the full range of confidence intervals is considered [30], suggesting a rapid, sequential splitting of the three subspecies. This scenario is expected to result in concordance factors that differ only slightly from 0.333, due to the short internal branch of the phylogeny. Our results are consistent with these patterns, with a primary phylogenetic history supported by a concordance factor of 0.390 across the autosomes. In contrast, three-taxon cases in *Drosophila* and primates feature phylogenies with longer internal branches, resulting in a greater proportion of the genome supporting the primary phylogenetic histories [17,22,23,25]. If incomplete lineage sorting is solely responsible for phylogenetic discordance, the two minor topologies should occur at equal frequencies in the genome [9,12,56,76], and these frequencies should decrease at equal rates as effective population size decreases and the length of the internal branch increases [11,15]. In contrast, our analysis revealed asymmetric genomic proportions supporting the two minor topologies, indicating a strong deviation from the model of pure lineage sorting. Similar patterns were observed in *Drosophila* species [17], and on the X chromosome in primates [10,22].

Gene flow following divergence can drive asymmetries between the minor histories. Patterns of shared polymorphism among populations [29–31] and introgression across hybrid zones [30,32– 39,74,75] indicate that gene flow differs among the subspecies pairs and across the genome. If the primary phylogenetic history (M. m. musculus/M. m. castaneus) represents the subspecies history, high levels of gene flow from the outgroup (M. m. domesticus) into M. m. castaneus or M. m. musculus could raise support for the minor histories. Significant levels of gene flow have only been detected between M. m. musculus and M. m. castaneus and between M. m. domesticus and M. m. castaneus [30]. This introgression is expected to increase support for the M. m. castaneus/M. m. domesticus minor history in respect to the M. m. musculus/M. m. domesticus minor history, as observed in our data.

In addition to gene flow in nature, sequencing error likely contributed to differences in concordance factors between the two minor histories. It has been suggested that sequencing errors could have caused differences in the genomic proportions supporting alternative minor histories in Drosophila and on the primate X chromosome [10]. Resequencing studies have detected a high false negative rate against M. m. castaneus specific SNPs in this data set [59]. This bias probably led us to underestimate the concordance factor for the M. m. musculus/M. m. domesticus topology. Although we cannot separate the contributions of recent gene flow and ascertainment bias to the asymmetry between minor histories in our analyses, ascertainment bias seems to have played a larger role in producing this pattern. If the asymmetry between minor histories was mostly due to gene flow, we would expect it to be less apparent on the X chromosome because recent introgression has been relatively reduced on the X chromosome [30,34-36,39,74,75]. In contrast, differences between minor histories were similar for the X chromosome (0.099) and the autosomes (0.117) when the full range of credibility intervals was considered, suggesting that gene flow was not the primary underlying cause. Furthermore, the asymmetry was still present after the rat sequence was randomized at each locus across chromosomes 18 and 19. Because the shuffling erased phylogenetic signal due to orthology, lowered support for the M. m. musculus/M. m. domesticus topology was apparently caused by a shorter M. m. castaneus branch. This result also supports the idea that ascertainment bias contributed to the difference in concordance factors between the two minor histories.

Although ascertainment bias appears to have affected the relative frequencies of the minor histories, it does not seem to have interfered with our identification of a primary phylogenetic history. Randomly removing *M. m. castaneus* informative sites in our simulation study did not alter the difference between the *M. m. musculus/M. m. castaneus* and *M. m. castaneus/M. m. domesticus* topologies; both decreased at equal and linear rates as *M. m. castaneus* informative sites were removed. Assuming the asymmetry was entirely driven by ascertainment bias, we adjusted the data according to the simulations by lowering the concordance factors of the *M. m. musculus/M. m. castaneus* and *M. m. castaneus/M. m. domesticus* topologies equally until the two minor histories had equal

concordance factors (*M. m. musculus/M. m. castaneus*: 0.349; *M. m. castaneus/M. m. domesticus*: 0.326; *M. m. musculus/M. m. domesticus*: 0.326). Although these concordance factors do not include any effect of gene flow (an unrealistic assumption), these rough estimates allowed us to calculate the length of the internal branch of the subspecies tree that would maximize the likelihood of our dataset under a model of pure lineage sorting [80]. Using an ancestral population size of 120,000 (an average value across the three pairwise subspecies comparisons) [30] and assuming one generation per year, the concordance factors are consistent with an internal branch length of only 5,520 generations (95% CI: 2,640–8,400 generations). The high level of phylogenetic discordance we observed suggests a rapid splitting of the three house mouse subspecies, consistent with close divergence times among the three subspecies estimated from large population samples [30].

#### A Phylogenetic History for House Mouse Subspecies

The *M. m. musculus/ M. m. castaneus* primary phylogenetic history has significantly higher support across the genome, indicating that it is an accurate reflection of the subspecies tree. Several additional lines of evidence support this conclusion. First, there is significantly higher support for this topology on the X chromosome (relative to the autosomes) where incomplete lineage sorting is expected to be reduced. If either of the minor histories were the true subspecies tree, rates of gene flow would need to be higher on the X chromosome than on the autosomes to explain the difference in concordance factors. However, gene flow on the X chromosome is considerably lower [30,32,33,36,38,39,74,75].

Second, we observed increased support for the M. m. musculus/ M. m. castaneus primary history at a hybrid male sterility locus. In species that experience gene flow after the initial development of reproductive isolation, loci underlying reproductive barriers might better reflect species history because discordance generated by gene flow is reduced in these regions [76,81]. Increased phylogenetic resolution of species history has been observed at loci associated with hybrid male sterility [82–84]. Within house mice, loci that affect hybrid male sterility have been mapped repeatedly to the X chromosome in crosses between M. m. musculus and M. m. domesticus [46,48,49]. As a preliminary examination of the association between reproductive isolation loci and phylogenetic history, we performed concordance analyses in sliding windows comprised of four contiguous loci across the X chromosome. We identified several adjacent regions supporting a M. m. musculus/M. m. castaneus topology with significantly higher concordance factors than the remainder of the X chromosome (p < 0.05; calculated by comparison to results using random subsets of four loci; Figure 6). The highest peak spanned a 1.55 Mb region and was supported by a concordance factor of 0.998 (p = 0.0196). This region matched the estimated location of the hybrid male sterility locus identified by Storchová et al. [46]. As might be predicted from the mapping results, concordance factors supporting the M. m. musculus/M. m. domesticus topology were also markedly reduced in these regions. We did not detect similar associations for other known hybrid sterility loci, including Hst1 on chromosome 17 [85]. However, the results from the X chromosome should motivate similar analyses across the entire genome once more information is available about the regions contributing to reproductive isolation in house mice.

In addition to informing speciation studies, the phylogenetic history of house mouse subspecies has important implications for mouse genetics. The classical inbred mouse strains widely used in genetic studies of disease and other phenotypes are descended – in unequal proportions – from the three subspecies examined here [86,87]. Analyses of the Perlegen sequences documented substantial genomic variation in relationships among the classical strains and attempted to attribute the ancestry of different genomic regions to M. m. domesticus, M. m. musculus, and M. m. castaneus [57,59]. Our results suggest that much of this phylogenetic variation likely reflects incomplete lineage sorting and differential introgression in wild mice. Genomic comparisons involving larger numbers of wild mice will be needed to interpret the patterns of genetic variation observed in the classical strains.

#### Methods

#### Sequence Data

Phylogenetic analyses were conducted using genome sequences from three wild-derived inbred strains: WSB/EiJ (*M. m. domesticus*; Maryland), PWD/PhJ (*M. m. musculus*; Czech Republic), and



**Figure 6. Sliding window analysis of discordance across the X chromosome.** Discordance within a four-locus sliding window was calculated across the X chromosome and is plotted as the midpoint position of each window. The *M. m. musculus/M. m. castaneus* topology (red) shows significantly higher regions along the X chromosome where known hybrid male sterility loci are mapped (*Hstx1*) [46]. The *M. m. musculus/M. m. domesticus* topology (blue) decreases within these regions. The entire *Hstx1* interval is indicated by the purple line, whereas the peak of this quantitative trait locus is indicated by the orange line. The black line indicates the chromosome-wide concordance factor for the *M. m. musculus/M. m. castaneus* topology (0.450).

doi:10.1371/journal.pgen.1000729.g006

CAST/EiJ (M. m. castaneus; Thailand). Although WSB originated in North America, M. m. domesticus from eastern North America are genetically similar to European M. m. domesticus that are closer to the ancestral subspecies range [88]. For each strain, a genome sequence was reconstructed using single nucleotide polymorphism (SNP) resequencing data generated by Perlegen Sciences [57]. This data set was obtained by using an oligonucleotide array designed with the C57BL/6J strain as a reference sequence. We identified 4,359,927 million assays within this dataset that were polymorphic in one of the three strains (WSB/EiJ, PWD/PhJ, or CAST/EiJ). For each assay, we replaced the C57BL/6J genotype with the strain-specific genotype in the mouse genome [58] using positional information from NCBI build 36. Any missing SNP data (N's) were also substituted in to the subspecies-specific genome sequence. The completed genome sequences consisted of SNPs at an average density of 2.35 SNPs/kb, surrounded by large tracks of C57BL/6J genome sequence, which were invariant between the three subspecies.

#### Whole-Genome Alignment

To root the phylogenetic analyses, the genome sequence from Rattus norvegicus [60] was used as an outgroup. Rat was chosen because it was the most closely related species to mouse with a whole-genome sequence available. The entire C57BL/6J genome was used in the alignment except for chromosome Y, as there is no rat chromosome Y sequence available. The NCBI build 36 mouse genome sequence was aligned to version 3.4 of the rat genome using a combination of the Mercator and MAVID programs [89]. Mercator was used to build a one-to-one colinear orthology map between the two genomes and MAVID was run on the resulting colinear blocks to produce nucleotide-level alignments. The input to Mercator consisted of all coding exon annotations for mouse and rat available from the UCSC Genome Browser [90] as well as the results from running BLAT [91] on the coding exon sequences in an all-vs-all fashion. Although coding regions formed the basis for the whole-genome alignment, all sequence data (coding and non-coding sequences) were utilized in subsequent phylogenetic analyses.

#### Minimum Description-Length Genome Partitioning

Orthologous blocks created from the alignment with rat were further partitioned into loci by measuring phylogenetic incongruence within the blocks using a minimum description length (MDL) principle [62]. Each orthologous block was split into consecutive 100 SNP intervals. Partitions were formed by combining these intervals into longer fragments. The Description Length (DL) of a partition was determined by the sum of the maximum parsimony tree lengths of each fragment in the partition, plus a penalty for the number of fragments. The penalty was set to be a constant cost (C) times the number of fragments. The partition with the smallest DL was selected. For example, consider the comparison between a partition with two adjacent fragments and a partition with the two fragments concatenated into a single large fragment. The penalty was 2\*C for the former partition and 1\*C for the latter. Concatenation was favored when the difference between the tree length of the two concatenated fragments and the sum of the tree lengths of the non-concatenated fragments was smaller than C (the penalty difference). Alternatively, two adjacent fragments were kept separate if the difference was greater than C, reflecting a shift in phylogenetic history of one region. The maximum and minimum costs for four taxa were used. The maximum cost favored fewer breakpoints whereas the minimum cost favored more breakpoints. A maximum cost of three was based on equation (2) in Ané and Sanderson [62]. A lower cost of 0.9039

was also used. A range of starting interval sizes was used (25, 50, 100, 150, 200, and 250 SNPs) on chromosomes 18 and 19 to determine the largest starting interval size that had similar concordance factors as the smaller interval sizes. This ensured the highest possible resolution while maintaining a reasonable computational time.

# Single-Locus Phylogenetic Analyses and Bayesian Concordance Analysis

Each locus identified from the MDL partitioning was subjected to a separate Bayesian phylogenetic analysis. Each locus was allowed to follow an independent model of molecular evolution, as determined by MrModelTest (Nylander 2004; http://www.abc. se/~nylander/mrmodeltest2/mrmodeltest2.html). The highest scoring model was selected based upon Akaike's information criterion (AIC) [92]. Each locus was subsequently analyzed using Mr.Bayes [93,94], with four Markov chains running for 2,000,000 generations (two simultaneous runs), discarding the initial 25% of the trees as burn-in. Topology and branch length priors were left at default settings. Convergence of runs was examined in a random subset of loci. Two million generations was sufficient to reach convergence in all examined cases. Each posterior distribution was used as input for a second-stage MCMC, Bayesian concordance analysis, using Bayesian Untangling of Concordance Knots (BUCKy) software [71] with 100,000 MCMC updates. The Bayesian concordance model incorporates a prior distribution of gene tree concordance ( $\alpha$ ). To examine the effects of variation in this prior, several values were compared: complete independence among loci ( $\alpha$  = infinity), a high probability of concordance among loci ( $\alpha = 10^{-6}$ ), and an intermediate level of concordance ( $\alpha = 1$ ). Analyses of locus size were performed for each topology by filtering out loci that did not have a posterior probability of at least 0.95, allowing unambiguous assignment of each locus to one of the three topologies. Locus size was defined with only the mouse sequences rather than including gaps introduced from the alignment with rat.

All phylogenetic analyses were repeated in a maximum likelihood (ML) framework to determine if inferences of the single-locus phylogenetic histories were robust to methodology. Each locus was allowed to follow an independent model of molecular evolution, as determined by ModelTest [95]. The highest scoring model was selected based upon Akaike's information criterion (AIC). ML searches were conducted using PAUP\* version 4.0b10 [96] using 500 bootstrap replicates to assess support. Because of the small number of taxa, heuristic search settings were left at default (TBR branch swapping, stepwise starting tree, simple taxon addition, 10 replicates). Rank correlations between posterior probability from Mr.Bayes and bootstrap support from PAUP\* were calculated for each topology using Spearman's rank correlation test. Concordance factors were calculated from the maximum likelihood trees by filtering out loci that did not have at least 95% bootstrap support and calculating the proportion of loci that supported a particular topology [23].

# Star Tree Paradox

Resolved gene trees can be supported by artificially high posterior probabilities when the actual tree is a hard polytomy or has a very short internal branch (the "star tree paradox") [66,68]. To determine whether the gene trees across the house mouse genome were resolved by a short internal branch or were actually polytomies, the maximum likelihood of the Bayesian majority rule consensus tree was calculated along with the maximum likelihood of the tree with an internal branch constrained to length zero. Both likelihood scores were calculated with PAUP\*, using identical model parameters summarized from the Mr.Bayes runs. The likelihood ratio test statistic was calculated as -2(-ln(restricted model) – ln(full model)). This test statistic was assumed to fit a mixed chi-square distribution where  $\alpha$  was twice the comparable value in a non-mixed, chi-square distribution of one degree of freedom (significance was set at  $\alpha = 0.1$ , 1 df,  $\chi^2 = 2.706$ ) [70].

#### Correlation with Recombination Rate

A sliding window analysis was used to compare locus size and recombination rate across the genome. Recombination rate was estimated within 1 Mb windows by linearly regressing genetic map position [64,65] on physical position of NCBI build 36 of the mouse genome. Windows were shifted at 250 kb intervals. The slope of this regression was used as an estimate of recombination rate (in cM/Mb) for the physical position at the midpoint of the window. If the window had less than three markers, it was discarded. The midpoint of each locus was paired with the closest estimate of recombination rate. Correlations were calculated using Spearman's rank correlation test. Statistical significance was estimated by permuting the recombination rate estimates while holding locus size constant, calculating the correlation, and repeating this permutation 100,000 times to generate a null distribution of correlation coefficients. Because the distance between a locus midpoint and the nearest estimate of recombination rate varied, correlation analyses were conducted by progressively restricting the data set from all data to only including loci within: 1 Mb, 500 kb, 2501kb, 100 kb, 75 kb, and 50 kb of a recombination rate estimate. This procedure tested whether the correlation became stronger when the analysis was restricted to loci associated with the closest estimates of recombination rates.

To ascertain whether there were statistical differences in recombination rate between loci with different sizes, the median recombination rate was determined for the 352 (2.5%) largest and smallest locus sizes. Statistical significance was estimated under the null hypothesis that the largest and smallest loci were actually from the same distribution. Both the largest and smallest loci were pooled. Random samples of loci were drawn with replacement from the combined pool to generate 352 large loci and 352 small loci test sets. The difference in median recombination rate between the two test pools was used as the test statistic. Sampling with replacement was repeated 100,000 times to generate a null distribution of test statistics.

# Long-Branch Attraction

To determine whether the phylogenetic discordance observed across gene trees was caused by long-branch attraction to one of the mouse lineages within the ingroup [72,73], we compounded any effect of long-branch attraction by randomizing the nucleotides of the rat sequence at every locus across chromosomes 18 and 19 [97]. This increased the length of the rat branch, erasing phylogenetic signal between rat and house mice. Each randomized locus was then subjected to Bayesian phylogenetic analysis and used as input for the Bayesian concordance analysis as described above with a prior of complete independence among loci ( $\alpha$  = infinity). The entire randomization test was repeated five times. A chi-square test was conducted on each replicate to determine whether the concordance factors of the three topologies differed from the 1/3, 1/3, 1/3 proportions expected if rat was randomly pairing with any of the three ingroup taxa. Only one replicate is reported, as each replicate had nearly identical results.

#### Ascertainment Bias Simulation

Yang et al. [59] documented an ascertainment bias in these data against SNPs that are consistent with a M. m. musculus/M. m.

domesticus topology. To understand the effects of this bias on our results, we conducted a series of simulations that mimicked varying degrees of bias in SNP discovery by artificially shortening the length of the M. m. castaneus branch. Ten percent to 80% of M. m. castaneus specific SNPs were randomly removed from the orthologous blocks generated by the whole-genome alignment with rat, resulting in eight separate concordance analyses with increasing levels of ascertainment bias. To remove the SNPs, a randomly selected M. m. castaneus genotype was changed to the corresponding C57BL/6J genotype at that position to simulate a false negative result on a high-density oligonucleotide array [57]. Each biased orthologous block was partitioned by MDL, subjected to Bayesian phylogenetic analysis, and used as input for the Bayesian concordance analysis as described above. For computational tractability, the ascertainment bias simulations and analyses were restricted to chromosomes 18 and 19, which exhibited different patterns of chromosome-wide concordance.

#### Supporting Information

**Figure S1** Distribution of locus sizes. Using a minimum description length principle, the genome was partitioned into 14,081 loci with a median size of 98,238 bp (SD 312,637 bp) and a maximum locus size of 7.21 Mb. Loci greater than 1 Mb in size are not shown.

Found at: doi:10.1371/journal.pgen.1000729.s001 (0.30 MB TIF)

**Figure S2** Recombination rate within large and small loci. The 2.5% largest loci (blue) have a significantly lower recombination rate as compared to the 2.5% smallest loci (red) (p<0.00001), suggesting the minimum description length principle partitioned the genome in a biologically informative manner.

Found at: doi:10.1371/journal.pgen.1000729.s002 (0.33 MB TIF)

**Figure S3** Single locus posterior probabilities. 84.9% of loci are supported by a high posterior probability (>0.9) from the single-locus Bayesian phylogenetic analyses, suggesting the minimum description length principle partitioned the genome in a phylogenetically informative manner.

Found at: doi:10.1371/journal.pgen.1000729.s003 (0.58 MB TIF)

**Figure S4** Fine-scale phylogenetic discordance. The posterior probability of each topology is mapped throughout the genome to characterize fine-scale patterns of discordance. Position along the chromosomes is indicated on the x-axis (Mb) and the posterior probability of each topology is on the y-axis. Colors correspond to the three topologies.

Found at: doi:10.1371/journal.pgen.1000729.s004 (6.81 MB TIF)

**Figure S5** Median locus size for each of the three topologies. Median locus size for each topology parallels the rank order of the concordance factors on both the autosomes and the X chromosome. Colors correspond to the three topologies.

Found at: doi:10.1371/journal.pgen.1000729.s005 (0.28 MB TIF)

**Figure S6** Maximum and minimum penalties against breakpoints for the minimum description length partitioning. Both the maximum (3) and minimum (0.9039) penalties were applied to the partitioning of chromosomes 18, 19, and X. Using a minimum penalty roughly doubles the number of loci on each chromosome, but the chromosome-wide concordance factors remain similar. Colors correspond to the three topologies. Error bars are 95% credibility intervals.

Found at: doi:10.1371/journal.pgen.1000729.s006 (0.32 MB TIF)

**Figure S7** Varied starting interval sizes for the minimum description length partitioning. A range of SNP intervals was applied to the partitioning of chromosomes 18 and 19. There are no

significant differences in the concordance factors between the first three starting intervals: 25, 50, or 100 SNPs. Colors correspond to the three topologies. Error bars are 95% credibility intervals.

Found at: doi:10.1371/journal.pgen.1000729.s007 (0.38 MB TIF)

**Figure S8** Phylogenetic discordance and long-branch attraction. The rat sequence was randomly shuffled to erase any phylogenetic signal between rat and house mice on chromosomes 18 and 19. Without the sequence shuffled (A), topologies significantly deviate from a 1/3, 1/3, 1/3 ratio. With the rat sequence shuffled (B), the topologies converge to a 1/3, 1/3, 1/3 ratio. This indicates the rat sequence provides a strong phylogenetic signal and the patterns of discordance are not driven by long-branch attraction. Colors correspond to the three topologies. Error bars are 95% credibility intervals.

Found at: doi:10.1371/journal.pgen.1000729.s008 (0.38 MB TIF)

**Table S1** Genomic locations and posterior probabilities of the 14,081 loci (computed with a prior probability of gene tree concordance set at  $\alpha = 1$ ).

#### References

- Rokas A, Williams BL, King N, Carroll SB (2003) Genome-scale approaches to resolving incongruence in molecular phylogenies. Nature 425: 798–804.
- Hackett SJ, Kimball RT, Reddy S, Bowie RC, Braun EL, et al. (2008) A phylogenomic study of birds reveals their evolutionary history. Science 320: 1763–1768.
- Helmkampf M, Bruchhaus I, Hausdorf B (2008) Phylogenomic analyses of lophophorates (brachiopods, phoronids and bryozoans) confirm the Lophotrochozoa concept. Proceedings Biological sciences/The Royal Society 275: 1927–1933.
- Kuo CH, Wares JP, Kissinger JC (2008) The Apicomplexan whole-genome phylogeny: an analysis of incongruence among gene trees. Molecular Biology and Evolution 25: 2689–2698.
- Zou X, Zhang F, Zhang J, Zang L, Tang L, et al. (2008) Analysis of 142 genes resolves the rapid diversification of the rice genus. Genome Biology 9: R49.
- Degnan J, Rosenberg N (2006) Discordance of species trees with their most likely gene trees. PLoS Genet 2: e68. doi:10.1371/journal.pgen.0020068.
- Kubatko LS, Degnan J (2007) Inconsistency of phylogenetic estimates from concatenated data under coalescence. Systematic Biology 56: 17–24.
- 8. Edwards S (2009) Is a new and general theory of molecular systematics emerging? Evolution 63: 1–19.
- Maddison WP (1997) Gene trees in species trees. Systematic Biology 46: 523–536.
- Slatkin M, Pollack JL (2008) Subdivision in an ancestral species creates asymmetry in gene trees. Molecular Biology and Evolution 25: 2241–2246.
- Tajima F (1983) Evolutionary relationship of DNA-sequences in finite populations. Genetics 105: 437–460.
- Pamilo P, Nei M (1988) Relationships between gene trees and species trees. Molecular Biology and Evolution 5: 568–583.
- Takahata N (1989) Gene genealogy in 3 related populations consistency probability between gene and population trees. Genetics 122: 957–966.
- Wu CI (1991) Inferences of species phylogeny in relation to segregation of ancient polymorphisms. Genetics 127: 429–435.
- Hudson RR (1992) Gene trees, species trees and the segregation of ancestral alleles. Genetics 131: 509–512.
- Moore WS (1995) Inferring phylogenies from mtDNA variation: mitochondrialgene trees versus nuclear-gene trees. Evolution 49: 718–726.
- Pollard DA, Iyer VN, Moses AM, Eisen MB (2006) Widespread discordance of gene trees with species tree in Drosophila: evidence for incomplete lineage sorting. PLoS Genet 2: e173. doi:10.1371/journal.pgen.0020173.
- Satta Y, Klein J, Takahata N (2000) DNA archives and our nearest relative: the trichotomy problem revisited. Molecular Phylogenetics and Evolution 14: 259–275.
- Chen FC, Li WH (2001) Genomic divergences between humans and other hominoids and the effective population size of the common ancestor of humans and chimpanzees. American Journal Of Human Genetics 68: 444–456.
- O'HUigin C, Satta Y, Takahata N, Klein J (2002) Contribution of homoplasy and of ancestral polymorphism to the evolution of genes in anthropoid primates. Molecular Biology and Evolution 19: 1501–1513.
- Wall JD (2003) Estimating ancestral population sizes and divergence times. Genetics 163: 395–404.
- Patterson N, Richter DJ, Gnerre S, Lander ES, Reich D (2006) Genetic evidence for complex speciation of humans and chimpanzees. Nature 441: 1103–1108.
- Ebersberger I, Galgoczy P, Taudien S, Taenzer S, Platzer M, et al. (2007) Mapping human genetic ancestry. Molecular Biology and Evolution 24: 2266–2276.
- 24. Hobolth A, Christensen OF, Mailund T, Schierup MH (2007) Genomic relationships and speciation times of human, chimpanzee, and gorilla inferred

Found at: doi:10.1371/journal.pgen.1000729.s009 (1.29 MB XLS)

# Acknowledgments

We thank M. Livny and Z. Miller for providing computational resources and assistance using the CONDOR distributed computing environment. We thank S. Smith for insightful advice during the course of this project and J. Good and A. Geraldes for helpful comments on the manuscript. We are also extremely grateful for thoughtful guidance provided by D. Baum.

# **Author Contributions**

Conceived and designed the experiments: MAW CA BAP. Performed the experiments: MAW CND. Analyzed the data: MAW BAP. Contributed reagents/materials/analysis tools: CA CND BRL. Wrote the paper: MAW BAP.

from a coalescent hidden Markov model. PLoS Genet 3: e7. doi:10.1371/journal.pgen.0030007.

- Burgess R, Yang Z (2008) Estimation of hominoid ancestral population sizes under bayesian coalescent models incorporating mutation rate variation and sequencing errors. Molecular Biology and Evolution 25: 1979–1994.
- She JX, Bonhomme F, Boursot P, Thaler L, Catzeflis F (1990) Molecular phylogenies in the genus *Mus*: comparative analysis of electrophoretic, scnDNA hybridization, and mtDNA RFLP data. Biological Journal Of The Linnean Society 41: 83–103.
- Boursot P, Din W, Anand R, Darviche D, Dod B, et al. (1996) Origin and radiation of the house mouse: Mitochondrial DNA phylogeny. Journal of Evolutionary Biology 9: 391–415.
- Suzuki H, Shimada T, Terashima M, Tsuchiya K, Aplin K (2004) Temporal, spatial, and ecological modes of evolution of Eurasian Mus based on mitochondrial and nuclear gene sequences. Molecular Phylogenetics and Evolution 33: 626–646.
- Salcedo T, Geraldes A, Nachman M (2007) Nucleotide variation in wild and inbred mice. Genetics 177: 2277–2291.
- Geraldes A, Basset P, Gibson B, Smith K, Harr B, et al. (2008) Inferring the history of speciation in house mice from autosomal, X-linked, Y-linked and mitochondrial genes. Molecular Ecology 17: 5349–5363.
- Baines JF, Harr B (2007) Reduced X-linked diversity in derived populations of house mice. Genetics 175: 1911–1921.
- Boursot P, Auffray JC, Brittondavidian J, Bonhomme F (1993) The evolution of house mice. Annual Review Of Ecology And Systematics 24: 119–152.
- Sage RD, Atchley WR, Capanna E (1993) House mice as models in systematic biology. Systematic Biology 42: 523–561.
- Payseur BA, Krenz JG, Nachman M (2004) Differential patterns of introgression across the X chromosome in a hybrid zone between two species of house mice. Evolution 58: 2064–2078.
- Dod B, Smadja C, Karn RC, Boursot P (2005) Testing for selection on the androgen-binding protein in the Danish mouse hybrid zone. Biological Journal Of The Linnean Society 84: 447–459.
- Macholán M, Munclinger P, Sugerková M, Dufková P, Bímová B, et al. (2007) Genetic analysis of autosomal and X-linked markers across a mouse hybrid zone. Evolution 61: 746–771.
- Macholán M, Baird SJ, Munclinger P, Dufková P, Bimová B, et al. (2008) Genetic conflict outweighs heterogametic incompatibility in the mouse hybrid zone? BMC Evolutionary Biology 8: 271.
- Teeter KC, Payseur BA, Harris LW, Bakewell MA, Thibodeau LM, et al. (2008) Genome-wide patterns of gene flow across a house mouse hybrid zone. Genome Research 18: 67–76.
- Munclinger P, Bozikova E, Sugerkova M, Pialek J, Macholan M (2002) Genetic variation in house mice (Mus, muridae, rodentia) from the Czech and Slovak republics. Folia Zoologica 51: 81–92.
- Yonekawa H, Gotoh O, Tagashira Y, Matsushima Y, Shi LI, et al. (1986) A hybrid origin of Japanese mice "Mus musculus molossinus". Current topics in microbiology and immunology 127: 62–67.
- Orth A, Adama T, Din W, Bonhomme F (1998) Natural hybridization of two subspecies of house mice, Musculus domesticus and Mus musculus castaneus, near Lake Casitas (California). Genome 41: 104–110.
- Yonekawa H, Moriwaki K, Gotoh O, Miyashita N, Matsushima Y, et al. (1988) Hybrid origin of Japanese mice "Mus musculus molossinus": evidence from restriction analysis of mitochondrial DNA. Molecular Biology and Evolution 5: 63–78.

- 43. Yonekawa H, Takahama S, Gotoh O, Miyashita N, Moriwaki K (1994) Genetic diversity and geographic distribution of Mus musculus subspecies based on the polymorphism of mitochondrial DNA. In: Moriwaki K, Shiroishi T, Yonekawa H, eds. Genetics in Wild Mice. Tokyo and Karger, Basel: Japan Scientific Societies Press. pp 25–40.
- Iványi P, Vojtísková M, Démant P, Micková M (1969) Genetic factors in the ninth linkage group influencing reproductive performance in male mice. Folia biologica 15: 401–421.
- Forejt J, Iványi P (1974) Genetic studies on male sterility of hybrids between laboratory and wild mice (Mus musculus L.). Genetical research 24: 189–206.
- Storchová R, Gregorová S, Buckiová D, Kyselová V, Divina P, et al. (2004) Genetic analysis of X-linked hybrid sterility in the house mouse. Mammalian Genome 15: 515–524.
- Britton-Davidian J, Fel-Clair F, Lopez J, Alibert P, Boursot P (2005) Postzygotic isolation between the two European subspecies of the house mouse: estimates from fertility patterns in wild and laboratory-bred hybrids. Biological Journal Of The Linnean Society 84: 379–393.
- Good J, Dean MD, Nachman M (2008) A complex genetic basis to X-linked hybrid male sterility between two species of house mice. Genetics 179: 2213–2228.
- Good J, Handel M, Nachman M (2008) Asymmetry and polymorphism of hybrid male sterility during the early stages of speciation in house mice. Evolution 62: 50–65.
- Prager EM, Tichy H, Sage RD (1996) Mitochondrial DNA sequence variation in the eastern house mouse, Mus musculus: comparison with other house mice and report of a 75-bp tandem repeat. Genetics 143: 427–446.
- Prager EM, Orrego C, Sage RD (1998) Genetic variation and phylogeography of central Asian and other house mice, including a major new mitochondrial lineage in Yemen. Genetics 150: 835–861.
- 52. Tucker PK, Lee BK, Eicher EM (1989) Y-chromosome evolution in the subgeneus Mus (genus Mus). Genetics 122: 169–179.
- Lundrigan B, Jansa S, Tucker P (2002) Phylogenetic relationships in the genus mus, based on paternally, maternally, and biparentally inherited characters. Systematic Biology 51: 410–431.
- Liu Y, Takahashi A, Kitano T, Koide T, Shiroishi T, et al. (2008) Mosaic genealogy of the Mus musculus genome revealed by 21 nuclear genes from its three subspecies. Genes & Genetic Systems 83: 77–88.
- Goios A, Pereira L, Bogue M, Macaulay V, Amorim A (2007) mtDNA phylogeny and evolution of laboratory mouse strains. Genome Research 17: 293–298.
- Baum DA (2007) Concordance trees, concordance factors, and the exploration of reticulate genealogy. Taxon 56: 417–426.
- Frazer K, Eskin E, Kang H, Bogue M, Hinds D, et al. (2007) A sequence-based variation map of 8.27 million SNPs in inbred mouse strains. Nature 448: 1050–1053.
- Mouse Genome Sequencing C, Waterston RH, Lindblad-Toh K, Birney E, Rogers J, et al. (2002) Initial sequencing and comparative analysis of the mouse genome. Nature 420: 520–562.
- Yang H, Bell T, Churchill G, Pardo-Manuel De Villena F (2007) On the subspecific origin of the laboratory mouse. Nature Genetics 39: 1100–1107.
- Gibbs RA, Weinstock GM, Metzker ML, Muzny DM, Sodergren EJ, et al. (2004) Genome sequence of the Brown Norway rat yields insights into mammalian evolution. Nature 428: 493–521.
- Adkins RM, Gelke EL, Rowe D, Honeycutt RL (2001) Molecular phylogeny and divergence time estimates for major rodent groups: evidence from multiple genes. Molecular Biology and Evolution 18: 777–791.
- Ane C, Sanderson MJ (2005) Missing the forest for the trees: Phylogenetic compression and its implications for inferring complex evolutionary histories. Systematic Biology 54: 146–157.
- Slatkin M, Pollack JL (2006) The concordance of gene trees and species trees at two linked loci. Genetics 172: 1979–1984.
- Shifman S, Bell J, Copley R, Taylor M, Williams R, et al. (2006) A highresolution single nucleotide polymorphism genetic map of the mouse genome. PLoS Biol 4: e395. doi:10.1371/journal.pbio.0040395.
- Cox A, Ackert-Bicknell C, Dumont BL, Ding Y, Tzenova Bell J, et al. (2009) A New Standard Genetic Map for the Mouse. Genetics.
- Suzuki Y, Glazko GV, Nei M (2002) Overcredibility of molecular phylogeneies obtained by Bayesian phylogenetics. Proceedings of the National Academy of Sciences of the United States of America 99: 16138–16143.
- Cummings MP, Handley SA, Myers DS, Reed DL, Rokas A, et al. (2003) Comparing bootstrap and posterior probability values in the four-taxon case. Systematic Biology 52: 477–487.

- Lewis PO, Holder MT, Holsinger KE (2005) Polytomies and Bayesian phylogenetic inference. Systematic Biology 54: 241–253.
- Yang Z, Rannala B (2005) Branch-length prior influences Bayesian posterior probability of phylogeny. Systematic Biology 54: 455–470.
- Ota R, Waddell PJ, Hasegawa M, Shimodaira H, Kishino H (2000) Appropriate likelihood ratio tests and marginal distributions for evolutionary tree models with constraints on parameters. Molecular Biology and Evolution 17: 798–803.
- Ane C, Larget B, Baum DA, Smith SD, Rokas A (2007) Bayesian estimation of concordance among gene trees. Molecular Biology and Evolution 24: 412–426.
- Felsenstein J (1978) Cases in which Parsimony or Compatibility Methods Will be Positively Misleading. Systematic Zoology 27: 401–410.
- 73. Bergsten J (2005) A review of long-branch attraction. Cladistics 21: 163–193.
- Tucker PK, Sage RD, Warner J, Wilson AC, Eicher EM (1992) Abrupt cline for sex-chromosomes in a hybrid zone between 2 species of mice. Evolution 46: 1146–1163.
- Dod B, Jermiin LS, Boursot P, Chapman VH, Nielsen JT, et al. (1993) Counterselection on sex-chromosomes in the *Mus musculus* European hybrid zone. Journal of Evolutionary Biology 6: 529–546.
- Rosenberg N (2002) The probability of topological concordance of gene trees and species trees. Theoretical Population Biology 61: 225–247.
- Jeffroy O, Brinkmann H, Delsuc F, Philippe H (2006) Phylogenomics: the beginning of incongruence? Trends in genetics: TIG 22: 225–231.
- Galtier N (2008) Dealing with incongruence in phylogenomic analyses. Philosophical transactions of the Royal Society of London Series B, Biological sciences.
- Koehler KE, Cherry JP, Lynn A, Hunt PA, Hassold TJ (2002) Genetic control of mammalian meiotic recombination. I. Variation in exchange frequencies among males from inbred mouse strains. Genetics 162: 297–306.
- Wakeley J (2009) Coalescent theory: an introduction. Greenwood Village, Colo.: Roberts & Co. Publishers. pp xii, 326.
- Wu CI (2001) The genic view of the process of speciation. Journal of Evolutionary Biology 14: 851–865.
- Ting CT, Tsaur SC, Wu CI (2000) The phylogeny of closely related species as revealed by the genealogy of a speciation gene, Odysseus. Proceedings of the National Academy of Sciences of the United States of America 97: 5313–5316.
- Machado CA, Hey J (2003) The causes of phylogenetic conflict in a classic Drosophila species group. Proceedings Of The Royal Society Of London Series B-Biological Sciences 270: 1193–1202.
- Dopman EB, Pérez L, Bogdanowicz SM, Harrison RG (2005) Consequences of reproductive barriers for genealogical discordance in the European corn borer. Proceedings of the National Academy of Sciences of the United States of America 102: 14706–14711.
- Mihola O, Trachtulec Z, Vlcek C, Schimenti JC, Forejt J (2009) A mouse speciation gene encodes a meiotic histone h3 methyltransferase. Science 323: 373–375.
- Silver LM (1995) Mouse genetics: concepts and applications. New York: Oxford University Press, pp xiii, 362.
- Wade C, Daly M (2005) Genetic variation in laboratory mice. Nature Genetics 37: 1175–1180.
- Selander RK, Hunt WG, Yang SY (1969) Protein Polymorphism and Genic Heterozygosity in Two European Subspecies of the House Mouse. Evolution 23: 379–390.
- Dewey CN (2007) Aligning multiple whole genomes with Mercator and MAVID. Methods in molecular biology (Clifton, NJ) 395: 221–236.
- Karolchik D, Baertsch R, Diekhans M, Furey TS, Hinrichs A, et al. (2003) The UCSC Genome Browser Database. Nucleic acids research 31: 51–54.
- Kent WJ (2002) BLAT-the BLAST-like alignment tool. Genome Research 12: 656–664.
- Posada D, Buckley TR (2004) Model selection and model averaging in phylogenetics: Advantages of akaike information criterion and Bayesian approaches over likelihood ratio tests. Systematic Biology 53: 793–808.
- Huelsenbeck JP, Ronquist F (2001) MRBAYES: Bayesian inference of phylogenetic trees. Bioinformatics 17: 754–755.
- Ronquist F, Huelsenbeck JP (2003) MrBayes 3: Bayesian phylogenetic inference under mixed models. Bioinformatics 19: 1572–1574.
- Posada D, Crandall KA (1998) MODELTEST: testing the model of DNA substitution. Bioinformatics 14: 817–818.
- Swofford D (2002) PAUP\*: Phylogenetic analysis using parsimony, beta version 4.0b10. Sunderland, Massachusetts: Sinauer Associates.
- Wheeler WC (1990) Nucleic-acid sequence phylogeny and random outgroups. Cladistics 6: 363–367.